Author's response to reviews

Title: Incarvine C suppresses proliferation and vasculogenic mimicry of hepatocellular carcinoma cells via targeting ROCK inhibition

Authors:

Ji-Gang Zhang (grissomzhang@hotmail.com)
Dan-Dan Zhang (rainbowdan0513@hotmail.com)
Xin Wu (wuxinkaoyan007@163.com)
Yu-Zhu Wang (yzwyy0506@126.com)
Sheng-Ying Gu (gsy15815@163.com)
Guan-Hua Zhu (zghlxqx@163.com)
Xiao-Yu Li (lixiaoyulxb@sina.com)
Qin Li (liqin0626@hotmail.com)
Gao-Lin Liu (gaolinliu@aliyun.com)

Version: 4 Date: 3 September 2015

Author's response to reviews: see over
Dear Editor,

Thank you for giving us an opportunity to revise and improve our manuscript entitled “Incarvine C suppresses proliferation and vasculogenic mimicry of hepatocellular carcinoma cells via targeting ROCK inhibition” by Ji-Gang Zhang, Dan-Dan Zhang, Xin Wu, Yu-Zhu Wang, Sheng-Ying Gu, Guan-Hua Zhu, Xiao-Yu Li, Qin Li and Gao-Lin Liu. (ID: MS: 7520514217467856)

We have made revision according to your instructions and 2 reviewers’ positive and constructive comments and suggestions. All the changes in the revised manuscript are in red color.

Response to Reviewer 1

Minor Essential Revisions

> Comment 1: The author’s response to my comments is satisfactory to me. Especially, the explanations in “response to comment 1” are clear enough, and I suggest that the authors should add the whole explanations in the response to comment 1 into Discussion part. It will make the readers understand why the different effect between IVC and Y27632 was observed.

Response: Thanks for your valuable advices. We have made a corresponding revision in the Discussion of our manuscript.

Response to Reviewer 2

No comment.

Finally, we would like to express our gratitude for the time you have spent on our
response and manuscript, and hope that the correction will meet with approval.

Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Prof. & Dr., Gao-Lin Liu

Sept 2, 2015

Addresses: Department of Clinical Pharmacy, Shanghai General Hospital, Shanghai Jiao Tong University School of medicine, No.100 Haining Road, Shanghai, 200080, China

E-mail: gaolinliu@aliyun.com